Author's response to reviews

Title: High prevalence of Helicobacter pylori Infection in Behcet's Disease

Authors:
Kamran B Lankarani (lankaran@sums.ac.ir)
Mohammad Reza Ravanbod (drravanbod@yahoo.com)
Elham Aflaki (aflakye@yahoo.com)
Mohammad Ali Nazarinia (nazariniam@gmail.com)
Akbar Rajaee (rajaeeakbar@yahoo.com)

Version: 4 Date: 6 March 2014

Author's response to reviews: see over
Dear Editor

Thanks for review of the manuscript number (MS: 1701997361035251) on “High prevalence of Helicobacter pylori Infection in Behcet's Disease”.

The revised manuscript was prepared according to the editorial comments In this revised form following points could be mentioned:

1) Abstract was corrected.
2) 'Introduction' heading, was inserted.
3) Author Contributions section was added.
4) Competing interests section was added.
5) Conclusions section was separated.
6) The manuscript was checked for spelling mistakes at our best capacity.
7) In discussion, the reason for lower prevalence of H.pylori infection in our series was explained.

I have revised the manuscript according to the comments on behalf of all co authors.

Best Regards

Kamran B Lankarani M.D
Professor of Internal Medicine
Health Policy Research Center
Shiraz University of Medical Sciences
Islamic Republic of Iran